Author's response to reviews

Title: Decrease social inequalities return-to-work: development and design of a randomised controlled trial among women with breast cancer.

Authors:

Clémence Vidor (clemence.vidor@univ-lille3.fr)
Ariane Leroyer (ariane.leroyer@univ-lille2.fr)
Véronique Christophe (veronique.christophe@univ-lille3.fr)
Mélanie Seillier (melanie.seillier@carsat-nordpicardie.fr)
Jérome Foncel (jerome.foncel@univ-lille3.fr)
Jacques Bonneterre (j-bonneterre@o-lambret.fr)
Sophie Fantoni (fanquin@wanadoo.fr)

Version: 2 Date: 7 October 2013

Author's response to reviews: see over
Le 07/10/2013

Dear Dr Solera,

Please find the change made of our manuscript, Decrease social inequalities return-to-work: development and design of a randomised controlled trial among women with breast cancer by Clémence Vidor, Ariane Leroyer, Véronique Christophe, Mélanie Seillier, Jérome Foncel, Jacques Bonneterre et Sophie Fantoni, which we would like to submit for publication as a study protocol in BMC Cancer.

Attached to the article, you will find the original of external financing and the approval of the ethics committee (pdf files). Translated versions of the relevant parts are added. doc files.

No manuscript are under submission with other journals based on this study protocol.
It is stated in the manuscript in the Method / Design > intervention asessment methods > care and randomisation of patients that the participant who accept sign a consent form.

Yours sincerly, Clémence Vidor, Ariane Leroyer, Véronique Christophe, Mélanie Seillier, Jérome Foncel, Jacques Bonneterre et Sophie Fantoni.

Corresponding author : Clémence Vidor
University Lille Nord de France – Université de Lille III, Unité de Recherche en sciences Cognitives et Affectives (URECA) EA 1059, BP 60149 F-59653 Villeneuve-d’Ascq Cedex, France. clemence.vidor@univ-lille3.fr